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Accelerator Status
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Accelerator Status

RF separator has had issues since this
weekend.

MCC planned access to address the RF
separator during day shift.
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Phase 3!

Permissions
CEBAF
Beam Director's Status Credited
Destination Approval Beam | Current Controls
Mode Limit Comment Expiration | Status

CW beam
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(IDL1D00) . oW 200;uA ;%g;:v 2018 06:00 -
dump.
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Hall A ) invoked. 07-May- o

s =
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target Beam
current 22.5
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Fig. 1 [03/23/2018 17:46:26]
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7.2 -=> MCC completed the ion chamber functionality test with good results: logbook-> https://logbooks.jlab.org/entry/3546497
Fig. 3 [03/23/2018 20:14:15]
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https://logbooks.jlab.org/entry/3546454
https://logbooks.jlab.org/entry/3546497

Phase 3

7.3 was postpone until Raster work was completed Monday morning

7.4 ->The beam was nice and centered. https://logbooks.jlab.org/entry/3546587

Bill Gunning completed work on the raster during our access Monday morning: https://logbooks.jlab.org/entry/3548452
“After inspection a bad fiber optic cable running from the BRM to the FSD card was found and replaced. Status went back toy green.
Replaced the other BRM boards fiber too since it looked very old too.”

https://logbooks.jlab.org/files/2018/03/3548455/FADC _Tritium_2404.pdf
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https://logbooks.jlab.org/entry/3546587
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Hall Status!!

Took Optics over the weekend!
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Plot by: Shuije Li



MARATHON

MARATHON has collected about 70% of their experimental Data!
Kin 16 (RHRS P0= 2.9GeV/c theta=36.1)

Good electron counts: Tritium -> 16.6k He3->17.4k D2->15.6k
Plan for the next few days:

Finish up positron contamination study.

Begin MARATHON physics for LHRS and continue RHRS kin16!



